Author's response to reviews

Title: The prevalence, pattern and clinical presentation of developmental dental hard-tissue anomalies in children with primary and mix dentition from Ile-Ife, Nigeria

Authors:

Dada O Temilola (tesy4jii2004@yahoo.com)
Morenike O Folayan (tovinukpong@yahoo.co.uk)
Olawumi A Fatusi (ofatusi@yahoo.com)
Nneka M Chukwumah (nnechukwumah@yahoo.com)
Nneka Onyejaka (nnekaonyejaka@yahoo.com)
Elizabeth O Oziegbe (elioziegbe@yahoo.com)
Titus A Oyedele (ayotitus4christ@gmail.com)
Kikelomo A Kolawole (kikelomokolawole@gmail.com)
Hakeem Agbaje (kimexy@gmail.com)

Version: 8 Date: 4 October 2014

Author's response to reviews: see over
Dear Editor,

Cover letter

We editorial correction pointed out has been effected. The conclusion section has been properly addressed in line 292. Also the funding source has properly been addressed in the acknowledgement section (line 303-304). The fund for the project was out-of-pocket expenditure and we are not sure if this should actually be included in the manuscript. We have however included this in response to the comments.

Thanks.

Oluwaseyi Dada Temilola

Obafemi Awolowo University Teaching Hospitals Complex, Ile-Ife

Email: tesy4jil2004@yahoo.com

Tel: +2348167258512